Author’s response to reviews

Title: Frequency of foot ulcers in people with type 2 diabetes, presenting to specialist diabetes clinic at a tertiary care hospital, Lahore, Pakistan”.

Authors:

Bilal Younus (bilalbin@gmail.com)

Adeela Shahid (dradeela71@gmail.com)

Rozina Arshad (rozina979@gmail.com)

Saima Khursheed (dr.cyma@yahoo.com)

Muhammad Ahmad (mlashari7@gmail.com)

Haroon Yousaf (doctharoon@gmail.com)

Version: 2 Date: 27 Jun 2018

Author’s response to reviews:

Reviewer 1

This is a much improved manuscript and the authors have addressed my major points.

People with diabetes should not be referred to as 'diabetics', please change.

Needs some language corrections before being published

RESPONSE TO COMMENTS:

'diabetics' word is replaced with “People with diabetes” throughout the manuscript.

Replaced at these places:

In Abstract

under “Methods heading” 1st line; page 1

under “Results heading” 5th line; page 1

In Background Section:

In 2nd paragraph; 1st line & 8th line; page 3
In 3rd paragraph; 12th line; page 4

In Methods Section:

In Data Collection; 1st line; page 5

Heading of Table 2; page 11

In Discussion Section:

1st paragraph; 1st line; page 14

In Conclusion Section:

1st & 5th line page 14

Language corrections also done